Author's response to reviews

Title: Debriefing and video-based feedback of oral clinical presentations reduce anxiety in medical students of ICU: a multicentre, prospective, randomised study

Authors:

Matthieu SCHMIDT (matthieuschmidt@yahoo.fr)
Yonathan FREUND (yonatman@gmail.com)
Mickael ALVES (mikaelalves@ymail.com)
Antoine MONSEL (antoine.monsel@psl.aphp.fr)
Vincent LABBE (vinclabbe@gmail.com)
Elsa DARNAL (elsadarnal@gmail.com)
Jonathan MESSIKA (jonath.msk@gmail.com)
Jerome BOKOBZA (bokobzajm38@gmail.com)
Thomas SIMILOWSKI (thomas.similowski@psl.ap-hop-paris.fr)
Alexandre DUGUET (alexandre.duguet@psl.aphp.fr)

Version: 2
Date: 11 February 2014

Author's response to reviews:

The Response to the reviewers is uploaded above